Author’s response to reviews

Title: The Association between Alcohol, Marijuana, Illegal Drug Use and Current Use of E-cigarette Among Youth and Young Adults in Canada: Results from Canadian Tobacco, Alcohol and Drugs Survey 2017

Authors:

Vrati Mehra (vratimehra@gmail.com)
Asvini Keethakumar (asvinikeetha@gmail.com)
Yvonne Bohr (bohry@yorku.ca)
Peri Abdullah (peri15@yorku.ca)
Hala Tamim (htamim@yorku.ca)

Version: 1 Date: 22 Aug 2019

Author’s response to reviews:

Dear Dr. Böckerman,

Thank you for your kind consideration of our paper. We would like to thank the reviewers for their suggestions and for helping us make this paper better.

Please find enclosed a copy of our revised manuscript titled “The Association between Alcohol, Marijuana, Illegal Drug Use and Current Use of E-cigarette Among Youth and Young Adults in Canada: Results from Canadian Tobacco, Alcohol and Drugs Survey 2017”.

August 15, 2019

Dear Dr. Böckerman,